Author’s response to reviews

Title: Awareness, use and associated factors of emergency contraceptive pills among women of reproductive age (15-49 years) in Tamale, Ghana

Authors:

Anthony Amalba (amalbaanthony@yahoo.com)
Victor Mogre (vmogre@uds.edu.gh)
Monica N.A Appiah (appiah.monica@yahoo.com)
Winnifred A Mumuni (ayisha.winny@yahoo.com)

Version: 2 Date: 6 May 2014

Author’s response to reviews:

Line numbers has been provided. Authors' details have been updated.